Author's response to reviews

Title: Hepatotoxicity induced by trastuzumab used for breast-cancer adjuvant therapy: a case report

Authors:

Kazuo Ishizuna (k.ishizuna@hotmail.co.jp)
Jun Ninomiya (jinomiyajinomiyagrap.rape.plala.or.jp)
Toshihisa Ogawa (toshi@dokkyomed.ac.jp)
Eiichi Tsuji (eiichi-tky@umin.ac.jp)

Version: 2 Date: 23 July 2014

Author's response to reviews:

1. Please remove the consent form in the additional files, as it is not required to include this during submission.

2. Please change the description of the patient’s gender to ‘female’. 

3. Please include an acknowledgement section at the end of the manuscript before the reference list. Please acknowledge anyone who contributed towards the study by making substantial contributions to conception, design, acquisition of data, or analysis and interpretation of data, or who was involved in drafting the manuscript or revising it critically for important intellectual content, but who does not meet the criteria for authorship. Please also include the source(s) of funding for all authors. Authors should obtain permission to acknowledge from all those mentioned in the Acknowledgements.

4. Please could you clarify if figure 1 is the table, as they should be included after the References list, together with the Figure legend section.

We modified the manuscript and submit a revised version.